Author's response to reviews

Title: Effect of an education programme for patients with osteoarthritis in primary care - a randomized controlled trial.

Authors:

Eva Ekvall Hansson (eva.ekvall-hansson@med.lu.se)
Malin Jönsson-Lundgren (malin.lundgren@skane.se)
Anne-Marie Ronnheden (anne-marie.ronnheden@skane.se)
Eva Sörensson (eva.sorensson@skane.se)
Åsa Bjärnung (asa.bjarnung@skane.se)
Leif E Dahlberg (leif.dahlberg@med.lu.se)

Version: 5 Date: 4 October 2010

Author's response to reviews: see over
BMC Musculoskeletal disorders

Dear Editor

Our manuscript *Effect of an education programme for patients with osteoarthritis in primary care – a randomized controlled trial* has once again been submitted online. We have made the following changes, according to the editor’s comments:

We have added a sentence in discussion about the reason for the lack of effect on self efficacy.
The sentence in the conclusion is changed as recommended.
The changes are highlighted through track changes in the manuscript.

We hope that you will find these modifications of the manuscript satisfactory for publication in *BMC Musculoskeletal disorders*

Yours sincerely

Eva Ekvall Hansson (corresponding author)
PhD, RPT
Lund University
Department of Clinical Sciences in Malmö/General Practice/Family Medicine
Malmö University Hospital
Entrance 72:28:11
SE-205 02 MALMÖ, Sweden.
telephone:+46 40 391358
fax:+46 40 391370
e-mail:eva.ekvall-hansson@med.lu.se